Author's response to reviews

Title: For money or service? A cross-sectional survey of preference for financial versus non-financial rural practice characteristics among Ghanaian medical students.

Authors:

Jennifer C Johnson (jennacj@umich.edu)
Emmanuel Nakua (manyorogh@yahoo.com)
Mawuli Dzodzomenyo (mdzodzo@hotmail.com)
Peter Agyei-Baffour (agyeibaffour@yahoo.co.uk)
Mawuli Gyakobo (mawuli_gyakobo@yahoo.com)
Kwesi Asabir (asabiruk@yahoo.co.uk)
Janet Kwansah (janetkwam@yahoo.co.uk)
S. Rani Kotha (rkotha@umich.edu)
Rachel C Snow (rcsnow@umich.edu)
Margaret E Kruk (mkruk@columbia.edu)

Version: 2 Date: 15 May 2011

Author's response to reviews:

Removed the "DRAFT" watermark over the text. My apologies for any confusion.